Author’s response to reviews

Title: Lower Vitamin D Levels in Saudi Pregnant Women are Associated with Higher Risk of Developing GDM

Authors:

Abdulrahman Al-AJlan (aalajl@hotmail.com)
Sara Almusharaf (sara.almosharruf@gmail.com)
Mona Fouda (monafoudaneel@yahoo.com)
Soundararajan Krishnaswamy (ksounder@gmail.com)
Naji Aljohani (najij@hotmail.com)
Amal Al-Serehi (aalserehi@kfmc.med.sa)
Eman Sheshah (eman_shesha@hotmail.com)
Naemah Alshingetti (alshingetti@yahoo.com)
Iqbah Turkistani (iqbalzmt@hotmail.com)
Afrah Alharbi (afalharbi@ksu.edu.sa)
Buthaynah Ali (buthinah.ali@gmail.com)
Aisha Ali (amali@kfmc.med.sa)
Gawaher Al-Saeed (ndaghri@ksu.edu.sa)
Nasser Al-Daghri (aldaghri2011@gmail.com)

Version: 5 Date: 19 Dec 2017

Author’s response to reviews:

Editor Comments:

1. Change in Authorship:

We note the addition of a new author, Kaiser Wani, since the original submission of the manuscript.
In line with COPE guidelines, BioMed Central requires written confirmation from all authors that they agree with any proposed changes in authorship of submitted manuscripts. We would therefore need to ask that all authors complete our change of authorship form, which can be downloaded from http://www.biomedcentral.com/submissions/editorial-policies#authorship.

Please read the instructions in the form carefully, and return the form within 14 days with all signatures (including from any newly proposed/removed co-author(s)) to BMCSeriesEditorial@biomedcentral.com. Please also see our guidelines regarding qualification for authorship (http://www.biomedcentral.com/submissions/editorial-policies#authorship).

Author Comments:

The authors thank the editor for the observations made. The full author list present in last revision including Kaiser Wani was there since the first submission of this manuscript dated January 8, 2017. Unfortunately, it got missed in the first revision. The error was rectified in the second revision done in October 2017. However, as suggested by respected Editor and in line with COPE guidelines, the required authorship change form is dully signed by all authors including Kaiser Wani, and submitted along with this revision as a separate scanned file named as “authorship+form_Scanned with signatures”.

2. Authors’ Contributions:

a. Currently, the contributions of Kaiser Wani (KAW) are missing from the AC section. The individual contributions of ALL authors to the manuscript should be specified in the Authors’ Contributions section. Guidance and criteria for authorship can be found here: http://www.biomedcentral.com/submissions/editorial-policies#authorship

Author Comments:

The authors thank the editor, however, the contributions of Kaiser Wani is listed in the Author’s contribution section in this version. The initials used are “KW”.

b. We note that initials have been included which do not correspond to any of the authors listed on the title page (MSA). Please clarify this discrepancy. Author Comments: The Authors full name and email address have been added as a table on the title page.
Author Comments:

The authors thank the editor for the observations made. This was unfortunately a typo error and has been rectified in this version.

3. Consent to Participate:

Please include a statement in the Ethics Approval and Consent to participate section indicating whether written informed consent was obtained from each participant. If you wish, you can copy the relevant statement from the methods.

Author Comments:

The authors thank the editor for the observations made. The required statement is added in “Ethics Approval and Consent to participate section” in the revised manuscript.